Author's response to reviews

Title: Trends in socioeconomic inequalities in smoking prevalence, consumption, initiation, and cessation between 2001 and 2008 in the Netherlands. Findings from a national population survey.

Authors:

Gera E Nagelhout (gnagelhout@stivoro.nl)
Dianne de Korte-de Boer (dianne.dekorte@maastrichtuniversity.nl)
Anton E Kunst (a.kunst@amc.uva.nl)
Regina M van der Meer (rvandermeer@stivoro.nl)
Hein de Vries (hein.devries@maastrichtuniversity.nl)
Boukje M van Gelder (boukje.van.gelder@rivm.nl)
Marc C Willemsen (mc.willemsen@stivoro.nl)

Version: 2 Date: 7 February 2012

Author's response to reviews:

We have added a statement that ethical approval was not required for our study.